Author's response to reviews

Title: Effectiveness of oral nutritional supplementation for older women after a fracture: a randomized controlled pilot study

Authors:

Ian D Cameron (ianc@mail.usyd.edu.au)
Susan E Kurrle (kurrle@bigpond.com)
Cesar Uy (cuy@nsccahs.health.nsw.gov.au)
Keri Lockwood (kalockwood@nsccahs.health.nsw.gov.au)
Lydia Au (Lydia_AU@Alexhosp.com.sg)
Frederieke G Schaafsma (f.g.schaafsma@gmail.com)

Version: 2 Date: 28 May 2010

Author's response to reviews:

Dr Norton,

The clinical trial number has now been added.

Sincerely,

Ian D Cameron